Author’s response to reviews

Title: Cannabis use and the risk of tuberculosis: A systematic review

Authors:

Clare French (clare.french@bristol.ac.uk)
Caroline Coope (caroline.coope@bristol.ac.uk)
Luke McGuiness (luke.mcguinness@bristol.ac.uk)
Charles Beck (charles.beck@phe.gov.uk)
Sophie Newitt (Sophie.Newitt@phe.gov.uk)
Lauren Ahyow (Lauren.Ahyow@phe.gov.uk)
Matt Hickman (Matthew.Hickman@bristol.ac.uk)
Isabel Oliver (Isabel.Oliver@phe.gov.uk)

Version: 2 Date: 05 Jun 2019

Author’s response to reviews:

Assistant Editor Comments:

1. Authors Contributions

-- Currently the contributions of CRM (Charles R Beck) and MH (Matt Hickman) are missing from the authors contributions section.

The specific contribution of CRB and MH has now been added to the Author contributions section (see page 18, lines 377-379).

2. Funding

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

The following sentence has been added at the end of the Funding section as follows "The funding body had no role in the design of the study, collection, analysis, or interpretation of data, or in writing the manuscript." (page 17, lines 367-368)
3. Formatting

Please change the Introduction to background.

This amendment has been completed (see page 4, line 51).

4. Clean Manuscript

At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files. Please ensure that all figures, tables and additional/supplementary files are cited within the text.

The revised manuscript and all figures and additional files have been cleaned and uploaded for the resubmission.